Author’s response to reviews

Title: Collectives of diagnostic biomarkers identify high risk subpopulations of haematuria patients: exploiting heterogeneity in large-scale biomarker data

Authors:

Frank Emmert-Streib (f.emmert-streib@qub.ac.uk)
Funso Abogunrin (abogs5@yahoo.com)
Ricardo de Matos Simoes (r.dematossimoes@qub.ac.uk)
Brian Duggan (bjuduggan71@hotmail.com)
Mark W Ruddock (mark.ruddock@randox.com)
Cherith N Reid (cherith.reid@randox.com)
Owen Roddy (oroddy02@qub.ac.uk)
Lisa White (lwhite11@qub.ac.uk)
Hugh F O’Kane (hugh_o_kane@hotmail.com)
Declan O’Rourke (Declan.ORourke@belfasttrust.hscni.net)
Neil Anderson (neil.anderson@belfasttrust.hscni.net)
Nambi Rajan (nambi.rajan@belfasttrust.hscni.net)
Kate Williamson (k.williamson@qub.ac.uk)

Version: 6 Date: 9 December 2012

Author’s response to reviews: see over
The BMC Series Editorial Production Team

8 December 2012

To whom it may concern

Thank you for the email re the major revisions required to our manuscript as detailed below

Author list: Frank Emmert-Streib, Funso Abogunrin, Ricardo de Matos Simoes, Brian Duggan, Mark W Ruddock, Cherith N Reid, Owen Roddy, Lisa White, Hugh F O’Kane, Declan O'Rourke, Neil Anderson, Nambi Rajan and Kate Williamson

Title: Collectives of diagnostic biomarkers identify high risk subpopulations of haematuria patients: exploiting heterogeneity in large-scale biomarker data

Journal: BMC Medicine

MS ID: 1271796137750173

We have changed the title to sentence case in the manuscript.

The author list is identical in the submission system and the manuscript, in both style and order.

We have revised the each author affiliation which now contains department, institution, street name and number, town, post code and country.

We have inserted a statement to say that the remaining authors have no competing interests.

We have removed all issue numbers from the reference list.

We have inserted titles for table 5-10 above each of the respective tables.

We have checked that all abbreviations used within a table or its legend are defined within that legend.

Please let me know if you have any queries or require any further information.

Yours sincerely,

Kate Williamson

Kate Williamson PhD
Senior Lecturer/PI Uro-oncology Group, Centre for Cancer Research and Cell Biology, Queen's University Belfast, 97 Lisburn Road, Belfast BT9 7BL Northern Ireland

T: +44(0)28 9097 2790
F: +44(0)28 9097 2776
Email: k.williamson@qub.ac.uk